Author's response to reviews

Title: Intestinal Obstruction: A Rare Complication Of Channeling Transurethral Resection Of The Prostate (TURP): a case report

Authors:

Ademola A Popoola Dr (ademola67@yahoo.com)
Kayode A Onawola Dr (onawolakayode@yahoo.com)
Michael D Adesina Dr (ademola67@yahoo.com)

Version: 3 Date: 26 September 2007

Author's response to reviews:

The Editor,
Journal of Medical Case Reports - BioMed Central

Dear Sir

Cover letter for revised manuscript

Intestinal Obstruction: A Rare Complication of Channelling Transurethral Resection of the Prostate (TURP): a case report

Kindly find the revised manuscript and other attached files.

The suggested corrections have been made:

¿ :a case report¿ has been added to end the title
¿ Tile has been removed from the second page
¿ Corrections in the grammar in the main text has also been made
¿ Additional subheadings have suggested have been added.
¿ Corrections have also been made concerning the figure legend

Yours sincerely,

Dr Ademola A. Popoola